Author's response to reviews

Title: Eating disorders - knowledge, attitudes, management and clinical experience of Norwegian dentists

Authors:

   Ann-Katrin Johansson (Ann-Katrin.Johansson@iko.uib.no)
   Anders Johansson (Anders.Johansson@iko.uib.no)
   Eva Nohlert (eva.nohlert@ltv.se)
   Claes Norring (claes.norring@ki.se)
   Anne Nordrehaug Åstrøm (Anne.Aastrom@iko.uib.no)
   Åke Tegelberg (ake.tegelberg@regionorebrolan.se)

Version: 6
Date: 1 October 2015

Author's response to reviews: see over
Dear Editor,

Editorial requests:

- Please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

Response: Informed consent from the participants was not required according to the response from the Ethical committee (REK Vest no. 2010/905-2) and the Norwegian Data Protection Office NSD (NSD no. 24040).

- In the Authors' Contributions section, please detail the specific contributions of each individual author, as specified in our Instructions for Authors

Response: All authors listed in this manuscript have made substantial contributions to this work. Ann-Katrin Johansson, Anders Johansson, Eva Nohlert and Åke Tegelberg: Planning and implementation of the project, data analysis/interpretation and manuscript drafting/finalization. Claes Norring, Anne Nordrehaug Åstrøm: Data analysis/interpretation and manuscript drafting/finalization.

Please find enclosed a new manuscript where our responses according to the editorial requests above have been inserted (highlighted in yellow).

Best regards

Ann-Katrin Johansson